Author's response to reviews

Title: C-reactive protein, established risk factors and social inequalities in cardiovascular disease -- the significance of absolute versus relative measures of disease

Authors:

Maria Rosvall (Maria.rosvall@med.lu.se)
Gunnar Engstrom (Gunnar.engstrom@med.lu.se)
Goran Berglund (Goran.berglund@med.lu.se)
Bo Hedblad (Bo.hedblad@med.lu.se)

Version: 10 Date: 13 May 2008

Author's response to reviews:

We have changed one word in the manuscript: instead of educational it now says socioeconomic on p.14, paragraph 1, line 4.